Author's response to reviews

Title: Angiotensinogen M235T gene variants and its association with essential hypertension and plasma renin activity in Malaysian subjects: A case control study

Authors:

Yee-How Say (sayh@hotmail.com)
King-Hwa Ling (khling@medic.upm.edu.my)
GnanaSothie Duraisamy (durai@medic.upm.edu.my)
Suzanne Isaac (rozita@medic.upm.edu.my)
Rozita Rosli (rozita@medic.upm.edu.my)

Version: 2 Date: 5 February 2005

Author's response to reviews:

Dear reviewer(s),

Recommended changes have been made and duly revised. Minor corrections throughout the text on grammar, title page, data and results were made. The following changes are major:

1) Addition of Table 1 and Table 2 (Previous Table 1 and Table 2 are now Table 3 and 4, respectively)

2) Newly improved (changes on labeling) Figure 1 and Figure 2

3) Addition of a paragraph in the Discussion section on the gender-associated AGT M235T prevalence.

All authors read and approved all the changes and revisions made onto the manuscript.

Thank you.